Author's response to reviews

Title: Interventions Encouraging The Use of Systematic Reviews by Health Policymakers and Managers: A Systematic Review

Authors:

Laure Perrier (l.perrier@utoronto.ca)
Kelly Mrklas (kmrklas@ucalgary.ca)
John N. Lavis (lavisj@mcmaster.ca)
Sharon E. Straus (sharon.straus@utoronto.ca)

Version: 5 Date: 11 April 2011

Author's response to reviews: see over
Dear Dr. Eccles and Dr. Mittman,

Thank you for considering our manuscript entitled:

**MS# 1432538755457839 Interventions Encouraging The Use of Systematic Reviews by Health Policymakers and Managers: A Systematic Review.**

We carefully reviewed the copy-edited version of the manuscript and have accepted all changes.

Sincerely,

Laure Perrier, Information Specialist
Li Ka Shing Knowledge Institute of St. Michael’s Hospital and University of Toronto
30 Bond Street
Toronto, ON M5B 1W8
l.perrier@utoronto.ca
+ 1 416 864-6060 (extension 77589)

**Back up contact:**
Sharon Straus
sharon.straus@utoronto.ca
+ 1 416 864-3068